Author’s response to reviews

Title: The Resonance® metallic ureteral stent in the treatment of malignant ureteral obstruction: A prospective observational study

Authors:

Jun Miyazaki (jmiyazaki@iuhw.ac.jp)
Mizuki Onozawa (mizukionozawa@iuhw.ac.jp)
Satoshi Takahashi (stakahas@sapmed.ac.jp)
Yuka Maekawa (yuka_mae@gifu-u.ac.jp)
Mitsuru Yasuda (super7@gifu-u.ac.jp)
Koichiro Wada (gmd17055@s.okadai.jp)
Yuji Maeda (y-maeda@mattohp.com)
Takuro Masaki (t-masaki@harasanshin.or.jp)
Akito Yamaguchi (yamaguchi@harasanshin.or.jp)
Masahiko Suzuki (Masahiko12o11u08@aol.com)
Yasuyuki Sakai (yasakai@cancer-c.pref.saitama.jp)
Tomokazu Kimura (tomokazu.kim@md.tsukuba.ac.jp)
Manabu Takai (manabu11gaku@yahoo.co.jp)
Kensaku Seike (k.seike-04.6.26@leaf.ocn.ne.jp)
Takahiko Hashimoto (th-uro@hyo-med.ac.jp)
Shingo Yamamoto (shingoy@hyo-med.ac.jp)

Version: 1 Date: 10 Jul 2019

Author’s response to reviews:

1. Textual overlap
   I corrected the paper according to the instructions of the editor.
   (page 15, line 252-259)(page 16-17, line 279-285)

2. Ethics approval and consent to participate
   I added in this section "Written informed consent was obtained from all participants".
3. Consent for publication
I revised in this section "Not applicable".

4. Funding
I added in this section "Funding body covered the costs of English proofreading and statistical analysis".

5. Authors' contributions
I changed the authors' name to full initials. I added in this section "Corresponding author confirmed that all the authors had read and approved the final version".